Author's response to reviews

Title: MMP28 (epilysin) as a novel promoter of invasion and metastasis in gastric cancer

Authors:

Pan Jian (panjian2008@163.com)  
Tao Yanfang (taoyanfang1982@163.com)  
Zhou Zhuan (zzhouzhuan@yahoo.com)  
Wang Jian (wj196312@vip.163.com)  
Zhu Xueming (zhuxueming@yahoo.com.cn)  
Ni Jian (Ni_jian2008@yahoo.com.cn)

Version: 4  Date: 23 May 2011

Author's response to reviews:

Department of Cell and Molecular Biology,  
Chinese Academy of Medical Sciences,  
Peking Union Medical College, Beijing, China, 100021  
Email addresses: ni_jian2008@163.com  
Tel: (86)0512-62956209  
Fax: (86)0512-62956209

Dear Editor,

Thank you for your suggesting of our manuscript (Ref No.9102197064654903). We have revised our manuscript and our point-by-point responses are provided below, and text that has been added or modified from the original text is shown in the revised manuscript. We hope that you will find it acceptable for publication in BMC cancer and we look forward to your response.

Sincerely,
Ni Jian

Department of Cell and Molecular Biology, Cancer Institute (Hospital), Chinese Academy of Medical Sciences, Peking Union Medical College, Beijing, China, 100021

Email addresses: ni_jian2008@163.com

Tel: (86)0512-62956209

Fax: (86)0512-62956209

Formatting changes to be made:

----------------------------------------

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)

1. Following the suggestion of editor, we formatted the author list.

Abstract: This should be identical in the manuscript and the submission system

2. Abstract has been formatted.

Authors’ contributions: Please use initials instead of full name.

3. Author’s contribution has been formatted.

Authors’ contributions: Please indicate that all authors read and approved the final manuscript.

4. Author’s contribution has been formatted.

Additional files: all additional files must be mentioned in the text in numerical order, or removed from the system
Additional files: Please ensure all files are referred to as additional files (not, for example, supplementary data)

5. additional files have been formatted.